



Article Does Mentoring Directly Improve Students' Research Skills? Examining the Role of Information Literacy and Competency Development

Anesito Cutillas ¹^[10], Eingilbert Benolirao ²^[10], Johannes Camasura ³^[10], Rodolfo Golbin, Jr. ⁴, Kafferine Yamagishi ^{5,6}^[10] and Lanndon Ocampo ^{6,*}^[10]

- ¹ College of Arts and Sciences, Argao Campus, Cebu Technological University, Cebu City 6021, Philippines; anesito.cutillas@ctu.edu.ph
- ² College of Technology and Engineering, Argao Campus, Cebu Technological University, Cebu City 6021, Philippines; eingilbert.benolirao@ctu.edu.ph
- ³ College of Computer, Information and Communication Technology, Cebu Technological University, Cebu City 6000, Philippines; johannes.camasura@ctu.edu.ph
- ⁴ College of Arts and Education, Moalboal Campus, Cebu Technological University, Cebu City 6032, Philippines; rodolfojr.golbin@ctu.edu.ph
- ⁵ Department of Tourism Management, Cebu Technological University, Cebu City 6000, Philippines; kafferineyamagishi@gmail.com
- ⁵ Center for Applied Mathematics and Operations Research, Cebu Technological University, Cebu City 6000, Philippines
- Correspondence: lanndonocampo@gmail.com

Abstract: This work proposes a structural model highlighting the research skills of undergraduate students. Due to the stages in their research project implementation, mentoring students becomes a crucial initiative in higher education institutions. Despite substantial progress in the literature linking mentoring and skills development, there is a lack of greater emphasis on research skills, especially for undergraduate students facing research work for the first time. Consequently, the direct relation between mentoring and research skills may not be straightforward. Thus, driven by social learning theory, the proposed model highlights the mediating effects of information literacy constructs and competency development on the relationship between mentoring and research skills. An empirical study of 539 participants via Partial Least Squares-Structural Equation Modeling supports six of the seven hypothesized paths. Three major theoretical contributions arise from the findings. Firstly, mentoring improves information-seeking skills and information-sharing behaviors and facilitates students' competency development due to the technical knowledge transfer from the faculty mentor to student mentees. Secondly, information literacy constructs and competency development promote research skills, emphasizing that students with those behaviors and capacities will achieve enhanced research skills. Finally, our findings suggest that mentoring does not directly translate to improved research skills; instead, information-seeking and sharing behaviors and competency development fully mediate such a link. Thus, mentors must shape these behaviors for mentoring to develop students' research skills. Theoretical and practical insights are outlined from these findings for university leadership to inform the design of mentoring initiatives for undergraduate students.

Keywords: mentoring; research skills; information seeking; information sharing; competence development; undergraduate research

1. Introduction

The active involvement of undergraduate students in scientific research yields numerous advantages for their academic and professional careers [1]. Engaging in research improves students' propensity to publish in high-impact research journals, experience professional fulfillment, and achieve greater economic and academic stability [2]. Barnett [3]



Citation: Cutillas, A.; Benolirao, E.; Camasura, J.; Golbin, R., Jr.; Yamagishi, K.; Ocampo, L. Does Mentoring Directly Improve Students' Research Skills? Examining the Role of Information Literacy and Competency Development. *Educ. Sci.* 2023, *13*, 694. https://doi.org/ 10.3390/educsci13070694

Academic Editors: Maria José Sá and Sandro Serpa

Received: 21 April 2023 Revised: 14 June 2023 Accepted: 22 June 2023 Published: 8 July 2023



Copyright: © 2023 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). argued that undergraduate research and inquiry are crucial for developing higher-order skills and academic dispositions that prepare students for the complexity of modern society and the economy. This growing consensus emphasizes the need for learning opportunities, prioritizing research and inquiry-based learning. Typically, undergraduate research projects involve multiple stages, such as conducting a literature review, devising a research plan, collecting and analyzing data, and presenting the results in a thesis manuscript, journal article, or conference proceeding. Throughout the research process, students typically receive various types of mentoring support, which could be formal or informal, and students can derive benefits from having multiple mentors who offer diverse forms of support [4]. Mentoring undergraduate research has been recognized as a highly effective practice that enhances the quality of education and learning in higher education [5,6]. Among those pivotal practices is the faculty-mentored undergraduate research experience, in which students collaborate with expert faculty mentors on discipline-based research activities [7]. However, mentoring extends beyond the formal faculty and undergraduate student relationships, which can involve various individuals (e.g., university staff, peers, family, and friends) and can occur in either one-on-one connections or small group settings, all of which contribute to student's educational success [8-10]. In the work of Mullen and Klimaitis [11], they differentiated mentoring from coaching, induction, or training. They emphasized that mentoring is an active process, not therapy, a unilateral endeavor, a panacea, a temporary fix, or a singular intervention. Some manifestations include providing students with written feedback on their thesis manuscripts to improve their academic writing [12]. Also, regular one-on-one consultations between mentors and students, including verbal feedback, result in successful thesis completion and higher student satisfaction [13]. During mentoring, a complex process of communication, collaboration, conflict, and understanding of mentor–student interactions occurs [5]. The complexity of the mentorship experience in undergraduate research is augmented by the students' emerging self-development [5].

Mentorship is a symbiotic relationship where the faculty mentor fosters the student's growth professionally and personally, offering opportunities for professional skills development, networking, and confidence [14]. Extensive research has been conducted over the past three decades on the benefits of mentorship [15]. However, despite significant milestones in the field, Crisp et al. [16] acknowledged that the definitions of mentorship had been a subject of debate and evolution over time, demonstrating the intricate nature of understanding and defining mentorship. Gershenfeld [17] found that the primary function of mentoring is academic support, followed by psychosocial/emotional support and role modeling. Mentoring aims to enhance skills and abilities, providing broad support and focusing on individual growth and achievement [18,19]. Studies in the domain literature found that mentoring can improve information-seeking skills, influence information-sharing behavior, aid in competency development, and improve research skills by providing guidance, support, and feedback [5,20–22]. Case [23] associated information seeking with the actions carried out to find information that meets the information requirements of individuals in their everyday undertakings. Lopatto [24] and Russell et al. [25] argued that mentorship in undergraduate research goes beyond faculty members, which suggests that other research staff, academic librarians, postdoctoral fellows, graduate students, and undergraduate peers play important technical or psychosocial roles in this process. For instance, Blaney et al. [26] highlighted the critical aspect of cascading mentorship, where senior students and staff, with postdoctoral fellows (or postdocs) as a case in point, mentor graduate students in laboratories. Their findings emphasized the hands-on support and troubleshooting students receive from postdocs, which significantly impact students' mental health. Chlomoudis et al. [27] observed that students seek information to improve their academic or professional performance. In consonance with other information and communication technologies, the internet has accelerated this process. Along this line espouses the criticality of information literacy as it involves identifying, retrieving, evaluating, and using information effectively, which are crucial for successful information-seeking skills. Research on information literacy and information seeking can help individuals develop strong

information-seeking skills essential for conducting high-quality research and achieving academic and professional goals [28].

Meanwhile, information sharing can improve research skills by enabling knowledge exchange, mutual learning, and collaboration. Mentees can gain fresh insights and diverse perspectives that can aid in refining research questions, creating more robust designs, and generating innovative and impactful findings. With effective and consistent mentoring initiatives, students develop varying degrees of competence. Competence is the ability to handle specific situations or tasks successfully. Its requirements will change with several factors, such as internationalization, new production methods, and the widespread implementation of information technology [29–31]. Competency development integrates skills, knowledge, and job attitude to achieve desired workplace behavior and performance and can be formal or informal, with the latter regularly occurring during work and characterized by a lack of focus on learning [32–35]. The relationship between undergraduate research and mentorship is an area that has received significant attention in academic literature. While there is some evidence to suggest that mentorship can positively impact undergraduate research skills, there is still a need for further exploration of this relationship, as previous findings were not too straightforward.

This study seeks to address the existing gap in the literature on mentoring by empirically investigating the influence of mentorship on the development of undergraduate students' research skills, using Bandura's social learning theory as a guiding framework. Bandura's social learning theory provides a theoretical lens for mentoring, suggesting that learning occurs in a social context through observation and modeling. Mentees look to their mentors as role models to develop new skills or gain knowledge from a mentoring relationship. The mentor demonstrates the skills and provides motivation and expectations for the mentee [5]. When the mentor is regarded as a role model, the mentee gains confidence, self-efficacy, and exemplary job performance [36]. While the mentoring approaches may differ, the main goal remains the same: for the mentor to act as a useful example for the mentee's progress. With insights into critical factors in developing research skills, this work explores the relationship between mentorship, information-seeking skills, information-sharing behavior, competency development, and research skills. The study aims to (1) develop an empirical model to demonstrate how mentorship contributes to the formation of information-seeking skills, information-sharing behavior, competency development, and research skills; (2) test the model with empirical data using Partial Least Squares–Structural Equation Modeling; and (3) examine the mediating effect of information literacy (i.e., information-seeking skills and information-sharing behavior) and competency development on the relationship between mentoring and research skills.

Furthermore, this study offers theoretical contributions. First, this work provides a better understanding of the mentorship's role in enhancing research skills by applying Bandura's social learning theory and testing its applicability to the model in explaining research skills among undergraduate students. This theoretical contribution advances our understanding of how mentorship can impact research skills. Second, this work offers insights into how mentorship can enhance undergraduate students' research skills. The findings of this work could provide insights into developing educational and training programs that aim to improve research capabilities among undergraduate students. Theoretical contributions of this work hold significance in developing countries, such as the Philippines, where the development of research skills among undergraduate students has become increasingly important. This study sheds light on the role of mentorship in developing research skills, which can help universities in the Philippines and other similar countries craft effective mentoring strategies for enhancing research skills among their students. Moreover, the practical implications of these theoretical contributions can be extended to universities worldwide. The insights provided by this study can guide university administrators in designing and implementing effective research mentoring programs that can enhance the research performance of their institutions. The knowledge gained from

this study can help universities build a pool of skilled researchers who can advance an understanding in their respective fields and positively impact society.

The rest of this paper is arranged as follows: Section 2 outlines the review of the relevant literature and hypothesis development that guides the proposed structural model. Section 3 describes the study methods, including sampling and data collection, the profile of the participants, and data analysis. Section 4 reports the results of the measurement and structural models. Section 5 discusses the study findings and their insights. It ends with some concluding remarks and pathways for future works in Section 6.

2. Literature Review and Hypotheses Development

2.1. Bandura's Social Learning Theory

Bandura's social learning theory highlights the influence of social interactions on shaping a person's characteristics, behaviors, and skills, emphasizing the role of performance feedback and modeling persuasion [37]. It was deemed the first to include "modeling" or "vicarious learning" as a form of social learning [38,39]. Social learning theory acts as a link between behaviorism, which focuses on modifying external behavior through reinforcement and repetition, and cognitive theory, emphasizing the cognitive aspects of learning beyond routine memorization [37]. Furthermore, self-education and self-development play a crucial role in this process, involving self-regulation and the ability to guide oneself, make choices, and navigate challenges, including moral conduct [40–42]. In the context of social learning theory, Edinyang [38] established the difference between "imitation" and "modeling". Imitation is the learner's capability to reproduce observed behavior, whereas modeling involves a more intricate process with four crucial steps to facilitate successful observational learning. Bandura [42] outlined a process by which people learn through modeling and observation: (1) observing behavior, (2) internalizing it through cognitive processes, (3) replicating it through personal actions, and (4) receiving positive feedback to enhance motivation and self-efficacy, leading to improved performance, especially for individuals with high self-efficacy. Bandura [41] asserted that the presence of these components is pivotal in influencing whether or not imitation occurs when exposed to a model.

2.2. Mentoring

In the context of undergraduate research projects, mentorship is a more bonded and symbiotic relationship in which the faculty mentor fosters student growth professionally and personally [14,43]. In a mentoring relationship, mentees consider the mentor a role model for developing new skills or gaining knowledge. The mentor demonstrates the skills and provides motivation and expectations for the mentee [5]. Mentoring offers professional skills development, networking, professional values and attitudes, and confidence [14,44]. It is considered a critical approach for developing research skills and promoting faculty success [20]. Symbiotically, mentoring is an efficient approach for emerging researchers to understand better how academics conduct research, enhance their research skills, and build confidence in advancing their research interests [21]. Although mentoring approaches may differ, the main goal is for the mentor to serve as a valuable role model for the mentee's growth. The primary objective of mentoring is to enhance skills and abilities, which requires accommodating various interpersonal styles and expectations [21]. Jacobi [18] identified three aspects of mentoring: (1) focus on individual growth and achievement, (2) broad forms of support (i.e., professional and career development), and (3) the personal and mutual nature of mentoring relationships. Luna and Cullen [45] pointed out that mentoring relationships could take various forms, such as informal or formal, short-lived or longterm, and can be either planned or spontaneous. Furthermore, Mullen and Klimaitis [11] provided an overview of the nine diverse classifications of mentoring alternatives derived from the empirical literature on educational mentoring. Recent works on mentoring focus on peer mentoring or cascading (see Blaney et al. [26]; Lorenzetti et al. [46]) and virtual or e-mentoring (see Tinoco-Giraldo et al. [19]).

In mentoring, a mentor can assist an individual in identifying and prioritizing their information needs and developing effective strategies for discovering and assessing information. A mentor can also provide feedback on the quality of the information, allowing the mentees to improve their information-seeking skills [20]. For instance, a mentor could introduce mentees to new sources of information and alternative research methods, broadening their understanding of the research process and assisting them in developing a more sophisticated approach to information seeking. Also, mentoring influences informationsharing behavior by creating a supportive environment where people feel comfortable sharing their knowledge and experiences with others [21]. Individuals may become more confident in their abilities and more willing to share their insights and perspectives with the help and encouragement of a mentor [20]. A mentor may also demonstrate the value of sharing knowledge and expertise with others by modeling positive information-sharing behaviors. Mentoring can help build more robust and collaborative students by fostering a culture of information sharing, where individuals can learn from one another and achieve their goals together. Also, mentoring supports competency development by providing guidance, support, and feedback as they gain new knowledge and skills [5,11]. A mentor helps identify areas in which mentees need to improve their competence and can provide resources and support to achieve their goals. For instance, mentors can provide valuable perspectives and guidance by sharing their own experiences and insights, assisting the individual to overcome challenges and progress in their development. This may boost the mentees' confidence and self-efficacy for continuous growth and improvement. Overall, mentoring can be an effective tool for promoting competency development and assisting individuals in reaching their full potential. Consequently, mentoring improves research skills by providing guidance, support, and feedback throughout the research process [22]. For instance, mentors offer feedback, identify areas for improvement, and provide encouragement and motivation.

Several studies emphasized the influence of mentoring on the performance of undergraduate students in the research context. Haege and Fresquez [47] explored the effects of mentoring strategies. They examined their impact on student success and academic skill development by focusing on a diverse population of students at a public, minority-serving institution. Palmer et al. [5] analyzed the intertwined relationships between mentoring, undergraduate research, and student identity development, focusing on the outcomes of mentored undergraduate research to better understand the relationships and develop a research agenda that informs the practice of mentoring undergraduate research in higher education. Furthermore, Agricolo et al. [48] analyzed mentors' diagnostic behavior by outlining various traits that mentors use to assess students' research skills. Nolan et al. [49] examined the benefits and barriers of undergraduate research experiences in statistics from the perspective of students, faculty mentors, and institutions. Thus, mentored research benefits diverse students and does not affect their timely graduation rates. The quality of mentoring, including socioemotional and culturally relevant mentoring, and the duration of the mentorship influence students' learning and development during research experiences [47]. Blaney et al. [26] focused on cascading mentorship, which sheds light on the distinct mentoring role of postdocs and explores the nature and potential outcomes of interactions between students and postdocs.

Nevertheless, several studies have explored the relationship between undergraduate research and mentorship. Some of these studies have found that mentorship can improve undergraduate research skills like critical thinking, problem solving, and communication. For example, Laursen et al. [50] discovered that undergraduate students with mentors reported higher research skills than those without. However, while the relationship seems straightforward, it is important to emphasize that the findings of several studies disagree on the relationship between mentorship and undergraduate research skills. For instance, Malcom and Feder [51] found that mentorship had little impact on the development of undergraduate research skills. This suggests that the relationship between undergraduate research and mentorship is complex and may be influenced by several factors, including

the quality of the mentorship, the student's prior research experience, and the specific research project. Thus, while some findings suggest that mentorship positively influences undergraduate research skills, there is still a need to investigate this relationship further to understand better the factors that influence the efficacy of mentorship and the specific strategies by which mentorship impacts undergraduate research skills. Exploring this relationship can be useful for promoting research skill development and assisting students in academic institutions achieve their research endeavors. With these, the following hypotheses are offered:

Hypothesis 1 (H1). *Mentoring positively and significantly affects students' informationseeking skills.*

Hypothesis 2 (H2). *Mentoring positively and significantly affects students' informationsharing behavior.*

Hypothesis 3 (H3). *Mentoring positively and significantly affects students' competency development.*

Hypothesis 4 (H4). *Mentoring positively and significantly affects students' research skills.*

2.3. Information-Seeking Skills and Research Skills

Students are often faced with challenges and difficulties throughout their university stay. In response, their help-seeking skills are considered a deliberate act of seeking assistance, guidance, or support [52-54]. Help-seeking among students involves actively seeking solutions from others to address their problems, acting proactively to avoid a potential failure, and enhancing independent learning, ultimately leading to improved academic achievement [55,56]. In many cases, help-seeking behavior involves information-seeking skills. For instance, when students seek help, they often require specific information to understand better and find appropriate solutions to address their needs more effectively. Chlomoudis et al. [27] observed that students seek information to fulfill their information requirements, which, in turn, enhances their academic or professional performance. Information need is an individual's recognition that their knowledge is inadequate to attain an objective, while information behaviors encompass all forms of intentional and unintentional information-seeking patterns [57,58]. Information literacy is identifying information needs, identifying and retrieving relevant information, and evaluating and using information effectively and efficiently [59]. Furthermore, information literacy is an individual's ability to identify the need for, search for, evaluate critically, and use the information to solve problems in diverse situations [60]. Research on information literacy is closely linked to the study of information-seeking behaviors. A considerable body of literature investigates how students seek information in the context of learning and information literacies (e.g., [60,61]). The advent of the internet has accelerated information seeking [62].

Accordingly, information literacy research focuses on expressing information-seeking skills and educational outcomes. Wilson [63] proposed a nested model that partitions information behavior research into distinct subfields. The primary field is information behavior, within which the study of information-seeking behavior examines individuals' diverse techniques to discover and obtain access to information sources. Within the broader domain of information-seeking behavior, information-seerching behavior is a subset that concentrates specifically on the interactions between computer-based information systems and information users [64]. In an academic setting, seeking information is an essential component of the tasks given to the students. Students are expected to actively engage in the information-gathering process to draw conclusions based on information from various sources [65]. The critical characteristics of information seeking are the availability of various alternatives and the student's responsibility to find the required information. In a learning context, information behavior related to tasks is frequently imposed to elicit expected learning outcomes [65]. Howlader and Islam [66] revealed that most undergrad-

uate students require academic- and employment-related information. As a result, they frequently visit the library to perform research work and study for competitive job exams. According to Singh et al. [62], how undergraduates search for information within an academic environment involves searching for, assessing, choosing, and utilizing information for educational and academic objectives. Thus, information-seeking skills are an important aspect of the research process, and individuals with strong information-seeking skills are better equipped to conduct high-quality research. Students are expected to identify and find relevant information from various sources to conduct effective research. This necessitates information-seeking skills such as defining research questions, developing search strategies, evaluating sources for relevance and reliability, and synthesizing and analyzing information. Thus, we propose the following hypothesis:

Hypothesis 5 (H5). *Information-seeking skills positively and significantly affect students' research skills.*

2.4. Information-Sharing Behavior and Research Skills

Moore [28] defined information sharing as the exchange of relevant and timely information, both formally and informally, among participants (e.g., students). Bălău and Utz [67] argued that due to the proliferation of online collaborative platforms, it is essential to understand and effectively manage the dissemination of information in the present-day knowledge-based economy. Cho et al. [68] aimed to clarify the process of sharing information and to explore the internal and external motivators that may encourage individuals to share information on Facebook. Information-sharing behavior influences research skills by providing access to a broader range of ideas, perspectives, and resources. For example, when students share information, they can exchange knowledge, mutual learning, and collaborative efforts to achieve shared objectives. This facilitates the students in gaining fresh insights and diverse perspectives that can aid in refining their research questions, creating more robust research designs, and generating innovative and impactful findings. Thus, the following hypothesis becomes relevant.

Hypothesis 6 (H6). *Information-sharing behavior positively and significantly affects students' research skills.*

2.5. Competency Development and Research Skills

Competence is the ability of an individual to successfully handle certain situations or complete a specific task or job [29]. Adler [30] and Brown et al. [31] highlighted that the aspects of competency will change and become more rigorous due to several factors, such as globalization, the advent of novel production techniques, the extensive use of information technology, and the increasing significance of knowledge-based production in different organizations. Competency development integrates skills, knowledge, and job attitude to achieve the desired workplace behavior and performance [32]. Forrier and Sels [33] defined competency development as the actions undertaken by the employee and the organization to preserve or improve the employee's functional-, learning-, and career-related competencies. Competence development in organizations can be divided into two categories: formal learning, which is planned and organized with certification, and informal learning, which occurs regularly during work and everyday life with a low degree of planning and organizing. While formal learning is popular, informal learning occurs while primarily performing another task and is characterized by a lack of focus on learning [34,35]. Analogously in an academic setting, students' competence in carrying out undergraduate research projects stems from formal and informal learning, with a strong emphasis on the former. Universities design programs and allocate resources to develop students' skills in identifying critical research questions, seeking information sources,

designing research methods, synthesizing research findings, and disseminating project results. In addition, mentors and fellow students within and outside the institution help students gain competencies, including desired attitudes, in performing research tasks. On the other hand, students' access to the internet via video-sharing, e-learning, and even social networking platforms provides vast opportunities for increased research competencies. Thus, we propose the following hypothesis:

Hypothesis 7 (H7). *Competency development positively and significantly affects students' research skills.*

The proposed structural model and its hypothesized relationships are shown in Figure 1.



Figure 1. The proposed structural model.

3. Methods

3.1. Sampling and Data Collection

The measurement items for each construct in this work were adopted from validated measures generated from previous works, as summarized in Appendix A. Mentoring (MEN) construct has eight measurement indicators, competency development (CD) has seven measurement indicators, information-sharing behavior (ISB) construct has four measurement indicators, information-seeking skills (ISS) construct has three measurement indicators, and research skills (RS) construct has eight measurement indicators. The survey instrument was refined by academic experts, and the wording relevant to research skills was revised. Furthermore, the survey instrument measures all the constructs using a 7-point Likert scale with measurement items of all constructs ranging from 7 as "strongly agree" to 1 as "strongly disagree" and a qualifier question of "Have you taken any research course?". Also, a consent form was attached to the survey questionnaire to ensure that participants were adequately informed and gave consent to participate in the study on a voluntary basis. The consent form allows the participants to withdraw from answering the study at any time if they feel uncomfortable with the questions. This work utilized a random sampling technique. The survey instrument was translated to Google Forms for convenience, efficient administration, and cost efficiency. The survey questionnaire was personally administered by enumerators with smart devices and an internet connection and was only disseminated online for three weeks, from 3 March 2023 to 24 March 2023. The participants were undergraduate college students that have taken their research course (e.g., undergraduate thesis writing) in the Philippines. There were 610 responses collected. Thirty-seven did not provide consent to answer the survey, and thirty-four had non-engaging responses, which

were eliminated from the analysis. Of the 610, only 539 were valid and used for the final analysis. There was no missing data in the survey since all questions were required to have a response.

3.2. Profile of the Participants

The majority of the undergraduate student participants had an academic major in education (26%), engineering and technology (24%), and tourism and hospitality (16%), as presented in Figure 2. In this work, the undergraduate students were identified primarily as female (73%). Since a qualifier question was incorporated in the survey questionnaire, all participants completed a research course culminating in an undergraduate thesis project. They were prompted to think about their experience during their respective research classes and the mentor they had during the implementation of their undergraduate research project. In the case of most universities in the Philippines, undergraduate students are required to form a group of three to five members drawn from the same research class. Each group is supervised by a faculty member acting as an adviser who is considered their mentor in completing an undergraduate research project, a requirement for the course. It is noteworthy that advisers or mentors of these projects possess comparable qualifications, including holding a master's degree in the discipline they mentor. Also, faculty members in most Philippine universities obtained their undergraduate and graduate qualifications from nearby universities, with an estimated ratio of 1:100 faculty members obtaining their graduate studies abroad. This scenario demonstrates the homogeneity in the professional roles and qualifications of all mentors in undergraduate research projects.



Figure 2. Academic major of the participants.

3.3. Data Analysis Results

Two structural equation modeling approaches are widely available: (1) Partial Least Squares–Structural Equation Modeling (PLS-SEM) and (2) Covariance-Based Structural Equation Modeling (CB-SEM). Specifically, PLS-SEM and CB-SEM differ in several aspects. First, in studies that aim at testing or confirming a theory, the appropriate method is CB-SEM, which focuses on the model fit, while PLS-SEM aims to maximize the covariance between latent variables for improved model interpretation [69,70]. Second, CB-SEM is restricted to reflective constructs, although some studies utilized formative measurements in the structural model leading to identification issues. In contrast, PLS-SEM can analyze research models that include both reflective and formative constructs, providing a more flexible approach [71,72]. Third, CB-SEM requires meeting assumptions related to data normality, independence, and uniformity, which can lead to inaccurate results if violated, while PLS-SEM is a more robust method for analyzing data with a non-normal distribution, using standardization techniques to align with the central limit theorem [70,73–75]. Finally, PLS-SEM can perform predictive analysis of dependent latent variables, while CB-SEM

4. Results

4.1. Measurement Model Assessment

study used the SmartPLS software version 4.0.8.5.

The parallel testing of the outer measurement model and the inner structural model, as well as the presence of both reflective and formative latent variables, is facilitated by PLS analysis [79]. Since the proposed model in this study includes reflective measures, the first criterion in evaluating the model is to examine the validity and reliability of the measures [80]. As per the assessment of the measurement model, all indicators are convergent and reliable, as shown in Table 1. Convergent validity examines how well individual items align within a construct and assesses the correlation between items measuring the same variable [81,82]. A construct is considered to have convergent validity when its Average Variance Extracted (AVE) value is at least 0.5 [83]. However, this study accepts a factor loading of 0.65 for each item. Those with an outer loading above 0.65 are regarded as acceptable, while those with a loading value of less than 0.65 are removed [71]. No measurement indicators are removed after calculations through the SmartPLS algorithm since all indicators reach the threshold value of 0.65. All constructs have the appropriate convergent validity ranging from 0.573 to 0.724. Reliability and validity testing is conducted using composite reliability (CR) aside from the traditional Cronbach's (α). CR is preferred in PLS-SEM as it considers varying indicator loadings and avoids underestimation issues associated with Cronbach's α [77]. Furthermore, the measurement items are all reliable, with all the constructs reaching above Cronbach's alpha (α) threshold value of 0.70, which is considered a reliable and acceptable index [84,85], and the composite reliability (CR) threshold value of 0.70 [80]. The Cronbach's alpha ranges from 0.808 to 0.911, while the CR values range from 0.810 to 0.944. These results indicate high-reliability values.

for determining structural models in highly complex domains [78]. In the analysis, this

Table 1. Measurement model assessment results.

	Convergent Validity		Construct Reliability			Convergent Validity		Construct Reliability	
	Loading	AVE	CR	α		Loading	α	CR	AVE
CD1	0.770	0.653	0.912	0.911	MEN1	0.723	0.573	0.894	0.893
CD2	0.841				MEN2	0.762			
CD3	0.852				MEN3	0.758			
CD4	0.787				MEN4	0.793			
CD5	0.792				MEN5	0.773			
CD6	0.799				MEN6	0.730			
CD7	0.813				MEN7	0.768			
ISB1	0.809	0.639	0.81	0.808	MEN8	0.746			
ISB2	0.676				RS1	0.751	0.717	0.944	0.943
ISB3	0.856				RS2	0.847			
ISB4	0.844				RS3	0.870			
ISS1	0.810	0.724	0.818	0.810	RS4	0.894			
ISS2	0.869				RS5	0.879			
ISS3	0.872				RS6	0.847			
					RS7	0.841			
					RS8	0.839			

Note: α = Cronbach's alpha; CR = composite reliability; AVE = average variance extracted; CD = competency development; ISB = information-sharing behavior; ISS = information-seeking skills; MEN = mentoring; RS = research skills.

The degree to which a construct is empirically distinct from other constructs is called discriminant validity. Fornell and Larcker's [83] criterion is a common method in assessing discriminant validity and typically reveals collinearity issues in the inner model. The AVE of the discriminant validity constructs is greater than the squared correlation of each latent variable [83]. The square roots of the AVE are bolded in Table 2, whereas non-bolded values represent the intercorrelation value between constructs. All off-diagonal values are less than the square roots of AVE, indicating that the Fornell and Larker condition is satisfied.

	Competency Development	Information- Sharing Behavior	Information- Seeking Skills	Mentoring	Research Skills
Competency development	0.808				
Information-sharing behavior	0.409	0.800			
Information-seeking skills	0.421	0.507	0.851		
Mentoring	0.671	0.369	0.379	0.757	
Research skills	0.332	0.386	0.422	0.260	0.847

Table 2. Fornell and Larcker results.

Note: Square root of AVE is shown on the diagonal of the matrix in bold; inter-construct correlation is shown off the diagonal.

The Standardized Root-Mean-Square Residual (SRMR) measures the average discrepancy between observed and expected correlations in a structural equation model. It provides an absolute assessment of model fit by indicating the magnitude of these discrepancies. A threshold value below 0.10 or 0.08 [86] generally indicates a good fit. Henseler et al. [87] introduced the SRMR as a useful goodness-of-fit measure in PLS-SEM, helping to identify and prevent model misspecification. In this study, the SRMR value is 0.054, lower than the threshold standard acceptable fit value of 0.08. The Normed Fit Index (NFI) is a metric used in PLS-SEM to assess the overall fit of the model to the data [88]. It compares the fit of the estimated model with that of a null or baseline model, considering the degrees of freedom. The NFI ranges from 0 to 1, with values closer to 1 indicating a better fit. It shows how accurately the model captures the observed covariances between the variables. In this work, the NFI value is 0.848, reflecting a moderate fit, an acceptable value. Thus, the research model fitness demonstrates an acceptable fit.

4.2. Structural Model Assessment

This work assesses the predictive power of the endogenous variables in the model [89]. The main criteria for evaluating the structural model using PLS-SEM are the strength of path coefficients, R^2 values (prediction power), and f^2 (effect size), as suggested by Hair et al. [80]. The coefficient of determination (R^2) measures the model's predictive accuracy. The combined influence of the exogenous variables on the endogenous variable(s) ranges from 0 to 1, with 1 corresponding to complete predictive accuracy. The acceptable R^2 of 0.75, 0.50, and 0.25 correspond, respectively, to substantial, moderate, and modest levels of prediction accuracy [72,74]. In this study, R^2 provides the predictive accuracy of the structural model, as shown in Figure 3. CD explains the highest variance with an R^2 value of 0.468 (46%). Furthermore, other constructs have a modest prediction accuracy, with RS, ISS, and ISB having R^2 of 0.255 (25%), 0.162 (16%), and 0.152 (15%), respectively. All of the hypotheses (H1, H2, H3, H5, H6, H7) are supported except for H4. These are summarized in Table 3 and visualized in Figure 3.



Figure 3. Structural model.

Hypotheses	β	t Values	p Values	Decision
H1: Mentoring \rightarrow Information-seeking skills	0.379	8.363	0.000 ***	Supported
H2: Mentoring \rightarrow Information-sharing behavior	0.369	9.672	0.000 ***	Supported
H3: Mentoring \rightarrow Competency development	0.671	12.798	0.000 ***	Supported
H4: Mentoring \rightarrow Research skills	-0.012	0.179	0.858 ^{ns}	Not Supported
H5: Information-seeking skills \rightarrow Research skills	0.265	5.086	0.000 ***	Supported
H6: Information-sharing behavior \rightarrow Research skills	0.196	3.556	0.000 ***	Supported
H7: Competency development \rightarrow Research skills	0.148	2.514	0.012 **	Supported

Note: *** *p* < 0.000; ** *p* < 0.001; ns not significant.

The effect size, as defined by Cohen [90], Cohen [91], and Kock [92], represents the magnitude of an effect irrespective of the sample size being examined. Using the PLS algorithm, the effect size (f^2) values are estimated, indicating 0.02 (minor), 0.15 (medium), and 0.35 (substantial) effects on the relationship between exogenous and endogenous constructs [80]. Furthermore, a value less than 0.02 implies no effect of exogenous constructs on an endogenous construct. In this study, the f^2 results show that MEN has a substantial effect on CD ($f^2 = 0.818$). Moreover, MEN has a medium effect on ISB ($f^2 = 0.158$) and ISS ($f^2 = 0.168$). Both ISB and ISS have a medium effect on RS with ($f^2 = 0.035$) and ($f^2 = 0.062$), respectively. However, CD has a negligible effect on RS ($f^2 = 0.014$), and MEN has no effect on RS ($f^2 = 0.000$). These results are consistent with the other findings of the study.

4.3. Mediating Effect

The mediating relationships in the structural model are shown in Figure 3. In this work, the mediator analysis procedure in the PLS of Zhao et al. [93] is followed as Hair et al. [94] suggested for PLS-SEM. The result shows that information-seeking skills, information-

sharing behavior, and competency development fully mediate mentoring to students' research skills since H4 is not supported.

5. Discussion and Insights

Drawn from Bandura's social learning theory, this study examines the proposed structural model that highlights how undergraduate students' research skills are improved by mentoring initiatives of faculty researchers in an academic institution. In addition, it investigates the mediating roles of information-seeking skills, information-sharing behavior, and competency development in the relationship between mentoring and research skills. The proposed structural model finds its motivation from existing studies emphasizing the role of mentoring in offering guidance, support, and feedback to students along the research process [20,22]. Due to the compelling factors associated with mentoring, as espoused by Malcom and Feder [51], information literacy and competency development constructs are hypothesized to mediate the direct relationship between mentoring and research skills. Such an attempt reflects the components of social learning theory, which suggest that behaviors of individuals are not only shaped through observational learning (i.e., mentoring or role modeling) but also require cognitive processes that they deem to perform and the need for reciprocal determinism and feedback mechanisms. The limited insights in the domain literature require empirical evidence to support the design of initiatives to improve undergraduate students' research skills. Seven hypothesized paths are examined. The cross-sectional empirical analysis demonstrated in this work supports six of the seven hypotheses, leaving the direct path from mentoring to research skills unsupported.

The findings show that mentoring enhances the information-seeking skills of undergraduate students (H1), supporting the findings of Ransdell et al. [20]. An important component in the completion of the undergraduate research project, particularly in its initial phase (e.g., review of the extant literature), is the guidance of the mentor on how to search for references effectively (e.g., journal articles, conference papers, books) to build up the background of the project. This includes a critical understanding of the available platforms to search for these references, the quality of information present, and the amount of information necessary to build up the arguments that serve as the groundwork for the project. Over time, in a constant one-on-one interaction between the faculty mentor and student mentees, such skills are refined, the behavior is developed, and the seeking of information becomes more efficient. This view supports the social learning theory in a more straightforward manner in such a way that the information-seeking skills of mentees are driven by observational learning from their mentors. This relationship extends to the information-sharing behavior of students, supporting H2. With mentors possessing higher digital literacy for efficient collaboration and information sharing among peers via online platforms, the mentoring initiative facilitates knowledge transfer to students. Particularly in an undergraduate research project that comprises two or more students as member proponents, student mentees would find it more beneficial to gain those information-sharing skills from their mentors, such as the use of cloud storage and real-time collaboration platforms (e.g., Google Docs, Microsoft Teams, Slack, Asana, Trello, Zoom, Figma, Miro). With more relevance during the COVID-19 lockdowns, such tools have become imperative to support efficient communication among project members. Thus, as the study findings suggest, mentoring promotes undergraduate students' information literacy.

The role of mentoring in espousing competency development is also empirically supported in this study (H3). The most straightforward tasks of good mentors are ensuring that student mentees receive sufficient guidance, develop critical thinking, enhance their knowledge base of the discipline, receive helpful feedback, and promote creativity. These initiatives improve the necessary competencies pivotal in completing the research project. Since student mentees look up to mentors as role models, these competencies are honed during the mentoring process until the student's potential eclipses their mentors'. This finding also supports the core concept of observational learning and role modeling that the social learning theory describes in shaping behaviors and skills. On the other hand, the hypothesized relationship between the information-seeking behavior and research skills (H5) of students is supported in this study. Those skills that can identify sources of information, evaluate the quality of information from these sources, and utilize this information to draw arguments augment several aspects of undergraduate research project implementation. These include (1) the use of bibliographic records; (2) the formulation of a scientific problem, research objectives, and research hypotheses; (3) the selection of the population, the sample, and appropriate type of sampling; (4) the selection, development, and application of methods, techniques, and instruments; and (5) the analysis and processing of information. Students who lack the behavior of seeking, organizing, and utilizing information would find it difficult to execute these aspects in the research process. Hence, a straightforward role of information-seeking behavior is deemed apparent in developing research skills. Our finding also suggests that the information-sharing behavior of students develops their research skills (H6). A plausible view of this relationship is the two-way effect that students experience during information sharing. When students share critical information they know, they always get feedback from their peers and even outside the institution. The feedback loop helps refine their knowledge, correct knowledge gaps, and eventually enhance the existing information they possess. The continuous feedback during information sharing promotes those research skills (i.e., searching for relevant sources, formulating the research questions, selecting the appropriate methodology, properly applying the methods, and analyzing the findings), interpreting the results, drawing conclusions, and writing the final report. On the other hand, the competency development of students also promotes their research skills (H7). Such a relationship is almost straightforward. Students who develop the competencies necessary for implementing research projects demonstrate their research skills more efficiently and effectively. These competencies form the baseline for students to carry out the skills needed to complete their projects. The findings of H5, H6, and H7 can be explained from the lens of social learning theory in the following manner. The presence of information literacy and competency development of undergraduate students reinforces their self-efficacy, which is a central concept in Bandura's social learning theory. As students gain knowledge (i.e., through information seeking and information sharing), skills, and competencies in carrying out a research project, their belief in their own abilities increases.

Finally, the most critical finding of this study is the unsupported relationship between mentoring and research skills (H5). This suggests that mentoring does not directly translate to developing the research skills of students, at least in carrying out their undergraduate research projects. In effect, mentoring can only develop those skills if students possess information-seeking skills and information-sharing behaviors and develop their required competencies, as implied in the full mediating roles of their constructs. This finding offers important insights. First, those role-modeling demonstrations in the form of sharing the history of the mentor's career, revealing personal experiences, and conveying respect have an insignificant impact on molding students' research skills. This implies that abstract motivational sharing could hardly translate to the actual development of students' research skills unless they enhance students' behaviors to seek and share information and promote competency development. Secondly, while mentoring allows the mentor to recommend specific strategies to accomplish the project and help finish certain assigned research tasks, without the information literacy and developed competence of students, it does not improve their research skills. This may be viewed as follows. For instance, the mentor may direct a student to implement a specific modeling methodology to analyze the causal relationships between variables in a research project. While such directives may complete the tasks at hand, when students' creativity to seek and share information and students' effort to consciously perform abstraction about the tasks to improve their competence are missing, they cannot replicate the use of such a methodology to understand another differently framed research question. Thus, mentors must not only focus on directing the mechanical aspects of doing the tasks associated with the method; instead, they must put emphasis on

shaping students' behaviors to think and figure out why such a method is necessary for a given problem and how to manipulate and extend the same method to address future research questions. Such behaviors demand greater information literacy and competence development. Third, while students receive opportunities to learn new skills during the mentoring process, these skills must be paired with the conscious effort to distill, interpret, and utilize information to translate those opportunities into developed research skills. These insights associated with the full mediation of information literacy and competence development between mentoring and research skills are deemed novel in the literature.

Furthermore, the insights of our empirical study espouse the social learning theory of Bandura [42] in the following areas. First, role modeling and mentoring (i.e., observational learning) shape certain skills that may require less sophistication, such as increasing information literacy and competency development. Information-seeking skills can be directly shaped by observational learning as the process of information seeking can be thought of as highly mechanical and procedural than cognitive. Similarly, individuals develop some aspects of competencies from their mentors as part of observational learning. Meanwhile, one aspect of mentoring forms reinforcement through rewards and punishments that individuals (or students) receive from their mentors, which enables them to imitate and perform those skills being rewarded. Also, the exchange process of sharing information among peers, as in the case of members in a project, supports one of the fundamental aspects of social learning theory known as reciprocal determinism [42]. It suggests that individuals do not only learn from their environment, but they also actively influence their environment through the information they learn from various sources, facilitating the social aspect of the theory. Second, in relation to the first, observational learning may not shape highly sophisticated skills. Like other complex skills, research skills require more dynamic cognitive skills that can distill, synthesize, and combine concepts in a creative fashion and are adaptive to changes in several factors related to study design, data and information processing, weather conditions, policy-driven nuances, and other environmental factors. Thus, characterizing specific skills primarily driven by observational learning becomes imperative. Third, enforcing self-efficacy in individuals to perform sophisticated skills can be highlighted as a direct implication of the full mediation effects of information literacy and competency development between mentoring and research skills. Emphasized as fundamental in social learning theory, self-efficacy influences whether an individual performs or imitates a specific skill or behavior. In our empirical work, observational learning through mentoring promotes information literacy and competency development, eventually enhancing self-efficacy. Increasing self-efficacy for highly sophisticated skills, such as carrying out research projects, is critical for augmenting those skills. Thus, observational learning and role modeling must be designed to increase self-efficacy in complex tasks.

6. Concluding Remarks

6.1. Summary

This work proposes and validates an empirical model motivated by the social learning theory that explains the development of research skills of undergraduate students. Due to the multiple stages associated with undergraduate research projects, mentoring between faculty researchers and students within the same or different institutions becomes apparent. While the current literature offers empirical support linking mentoring and general research skills, specifically exploring undergraduate students' research skills is limited. Such an agenda may not be straightforward, as these students typically face research tasks for the first time. Thus, the proposed structural model espouses the possible mediating role of information-seeking skills, information-sharing behavior, and competency development in the relationship between mentoring and research skills. The model proposes seven hypothesized paths, and 539 valid participants are used to test these paths via PLS-SEM. All hypotheses are supported, except for the direct link between mentoring and research skills, implying that information literacy constructs and the competency development of students fully mediate such a relationship.

6.2. Implications

The findings of this study show three critical contributions. First, mentoring positively supports information-seeking skills, information-sharing behaviors, and competency development, partially supporting others in the literature. This implies that the technical guidance of the mentor on continuous one-on-one interaction with the student mentee refines the student's behavior in critically seeking information, assessing its quality, and creatively utilizing such information to build arguments for the research project, contest theories, and identify appropriate methodologies. On a similar note, mentors possessing high digital skills can transfer knowledge to students by sharing information to achieve collective success for the project, especially for undergraduate students who typically work in teams. The path from mentoring to competency development offers direct evidence regarding how mentors shape the required competencies of students to carry out research tasks. Secondly, the paths from information literacy and competency development constructs to research skills are positively supported in this study. This finding suggests that students who seek information and share it with peers or team members improve their research skills in accessing relevant sources, establishing research questions, evaluating appropriate methods, implementing those methods, interpreting findings, drawing relevant conclusions, and writing reports. In particular, the feedback loops in information sharing promote the development of the required research skills in completing projects. Finally, our findings demonstrate the full mediation effects of information literacy constructs and competency development, opposing the direct link of mentoring to research skills. This novel empirical support highlights the need for mentors to shape students' information-seeking and information-sharing behaviors to effectively translate mentoring efforts to developing hard research skills. They must focus beyond the mechanical aspects of performing the tasks to espousing students' creativity to consciously perform abstractions to address future research questions that may be framed differently. Such an initiative requires students to develop information-seeking and sharing behaviors, along with training that enhances their competencies.

6.3. Limitations and Future Work

Although these findings are pivotal in the literature, some limitations are evident. First, the participants in this study are from the Philippines, with inherent cultural and political makeups. Future work may expand our empirical model to a multi-cultural investigation with more participants. A multi-group analysis in PLS-SEM may identify the differences brought about by these varying cultural biases. Secondly, the participants belong to institutions with heterogeneous academic reputations. Some of them came from top-ranked universities with excellent faculty resources. In effect, the quality of the mentorship programs present in these universities may be significantly different from those in the lower-ranking quartiles. Thus, future studies may look into the technical competence of mentors as a moderating variable in developing research skills. Finally, predictive machine learning algorithms based on several factors leading to improving research skills may be interesting for future work.

Author Contributions: Conceptualization, K.Y. and L.O.; methodology, K.Y.; software, K.Y.; validation, K.Y.; formal analysis, K.Y. and L.O.; investigation, K.Y. and L.O.; resources, A.C., E.B., J.C. and R.G.J.; data curation, A.C., E.B., J.C., R.G.J. and K.Y.; writing—original draft preparation, A.C., E.B., J.C., R.G.J., K.Y. and L.O.; writing—review and editing, K.Y. and L.O.; visualization, K.Y.; supervision, A.C. and L.O.; project administration, A.C.; funding acquisition, A.C., E.B., J.C. and R.G.J. All authors have read and agreed to the published version of the manuscript.

Funding: The authors are grateful to the Office of the Vice-President for Research and Development of Cebu Technological University for partially funding this project.

Informed Consent Statement: Informed consent was obtained from all subjects involved in the study.

Data Availability Statement: Data will be made available upon request.

Conflicts of Interest: The authors declare no conflict of interest.

Appendix A

 Table A1. Measurement indicators.

Constructs	Indicators	References
Mentoring (MEN)	MEN1. My mentor shares the history of his/her career with me. MEN2. My mentor discusses my questions regarding feelings of competence and commitment to advancement. MEN3. My mentor shares personal experiences as an alternative perspective on my problems. MEN4. My mentor suggests specific strategies for accomplishing work objectives. MEN5. My mentor gives me assignments that present opportunities to learn new skills. MEN6. My mentor helps me to finish assignments. MEN7. My mentor conveys feelings of respect for me as an individual. MEN8. I respect and admire my mentor.	Younas and Bari [95]
Competency development (CD)	CD1. A more experienced mentor who guides me in my research and from whose experience I can learn. CD2. My adviser ensures I learn about the research by giving me challenging assignments. CD3. My adviser makes sure that I develop the competencies that I need for my research career. CD4. In my organization, training sessions are organized to gain knowledge. CD5. My adviser regularly gives me feedback about my performance. CD6. I can make use of personal development plans to know what competencies I need to develop. CD7. I have been given tasks that develop my competencies for the future.	Younas and Bari [95]

Constructs	Indicators	References
Information-seeking skills (ISS)	ISS1. I can use a variety of available options to search for information that my colleagues are not aware of. 1SS2. I can inform my classmates of different ways to effectively search for information. 1SS3. I can generate keywords to search for information for academic work.	Hong and Kim [96]
Information-sharing behavior (ISB)	ISB1. I can interact with classmates using real-time communication tools, for example, video conferencing tools or messengers. ISB2. I can share my opinions online, for example, with blogs, social networking services, or web pages. ISB3. I can share my files with classmates using online software. ISB4. I can collaborate with classmates using online software.	Hong and Kim [96]
Research skills (RS)	RS1. Use of catalogs, descriptor books and bibliographic records. RS2. Relation to the formulation of a scientific problem, research objectives, and research hypotheses. RS3. Selection of the population, the sample, and the type of sampling to be used. RS4. Selection, development, and application of methods, techniques, and instruments. RS5. Analysis and processing of information through different statistical techniques. RS6. Interpretation and discussion of results are presented in tables and graphs. RS7. Drawing up conclusions and recommendations. RS8. Writing final research reports.	Ipanaqué-Zapata et al. [97]

Table A1. Cont.

References

- Mass-Hernández, L.M.; Acevedo-Aguilar, L.M.; Lozada-Martínez, I.D.; Osorio-Agudelo, L.S.; Maya-Betancourth, J.G.E.M.; Paz-Echeverry, O.A.; Rahman, S. Undergraduate research in medicine: A summary of the evidence on problems, solutions and outcomes. *Ann. Med. Surg.* 2022, 74, 103280. [CrossRef] [PubMed]
- 2. Atallah, Á.N. Evidence-based medicine. *Sao Paulo Med. J.* **2018**, *136*, 99–100. [CrossRef] [PubMed]
- 3. Barnett, R. (Ed.) *Reshaping the University: New Relationships between Research, Scholarship and Teaching;* McGraw-Hill Education: Berkshire, UK, 2005.
- Lunsford, L.G.; Crisp, G.; Dolan, E.L.; Wuetherick, B. Mentoring in higher education. In *The SAGE Handbook of Mentoring*; Clutterbuck, D.A., Kochan, F.K., Lunsford, L., Dominguez, N., Haddock-Millar, J., Eds.; SAGE Publications Ltd.: Thousand Oaks, CA, USA, 2017; Volume 20, pp. 316–334.

- 5. Palmer, R.J.; Hunt, A.N.; Neal, M.; Wuetherick, B. Mentoring, undergraduate research, and identity development: A conceptual review and research agenda. *Mentor. Tutoring Partnersh. Learn.* **2015**, *23*, 411–426. [CrossRef]
- Linn, M.C.; Palmer, E.; Baranger, A.; Gerard, E.; Stone, E. Undergraduate research experiences: Impacts and opportunities. *Science* 2015, 347, 1261757. [CrossRef]
- Pierszalowski, S.; Bouwma-Gearhart, J.; Marlow, L. A systematic review of barriers to accessing undergraduate research for STEM students: Problematizing under-researched factors for students of color. *Soc. Sci.* 2021, 10, 328. [CrossRef]
- Erickson, L.D.; McDonald, S.; Elder, G.J. Informal mentors and education: Complementary or compensatory resources? *Sociol. Educ.* 2009, *82*, 344–367. [CrossRef] [PubMed]
- 9. Crisp, G.; Cruz, I. Mentoring College Students: A Critical Review of the Literature Between 1990 and 2007. *Res. High. Educ.* 2009, 50, 525–545. [CrossRef]
- 10. Marshall, M.; Dobbs-Oates, J.; Kunberger, T.; Greene, J. The peer mentor experience: Benefits and challenges in undergraduate programs. *Mentor. Tutoring Partnersh. Learn.* **2021**, *29*, 89–109. [CrossRef]
- 11. Mullen, C.A.; Klimaitis, C.C. Defining mentoring: A literature review of issues, types, and applications. *Ann. New York Acad. Sci.* **2021**, *1483*, 19–35. [CrossRef]
- 12. Basturkmen, H.; East, M.; Bitchener, J. Supervisors' on-script feedback comments on drafts of dissertations: Socialising students into the academic discourse community. *Teach. High. Educ.* **2014**, *19*, 432–445. [CrossRef]
- 13. Shanahan, J.O.; Ackley-Holbrook, E.; Hall, E.; Stewart, K.; Walkington, H. Ten salient practices of undergraduate research mentors: A review of the literature. *Mentor. Tutoring Partnersh. Learn.* **2015**, *23*, 359–376. [CrossRef]
- Johnson, W.B. Student-faculty mentorship outcomes. In *Blackwell Handbook of Mentoring: A Multiple Perspectives Approach*; Allen, T., Ebby, L., Eds.; Blackwell: London, UK, 2007; pp. 189–210.
- 15. Hamilton, L.K.; Boman, J.; Rubin, H.; Sahota, B.K. Examining the impact of a university mentorship program on student outcomes. *Int. J. Mentor. Coach. Educ.* **2019**, *8*, 19–36. [CrossRef]
- Crisp, G.; Baker, V.L.; Griffin, K.A.; Lunsford, L.G.; Pifer, M.J. *Mentoring Undergraduate Students*; ASHE Higher Education Report; Wiley Publishers: Hoboken, NJ, USA, 2017; Volume 43.
- 17. Gershenfeld, S. A review of undergraduate mentoring programs. Rev. Educ. Res. 2014, 84, 365–391. [CrossRef]
- 18. Jacobi, M. Mentoring and undergraduate academic success: A literature review. Rev. Educ. Res. 1991, 61, 505–532. [CrossRef]
- 19. Tinoco-Giraldo, H.; Torrecilla Sanchez, E.M.; García-Peñalvo, F.J. E-mentoring in higher education: A structured literature review and implications for future research. *Sustainability* **2020**, *12*, 4344. [CrossRef]
- Ransdell, L.B.; Lane, T.S.; Schwartz, A.L.; Wayment, H.A.; Baldwin, J.A. Mentoring new and early-stage investigators and underrepresented minority faculty for research success in health-related fields: An integrative literature review (2010–2020). *Int. J. Environ. Res. Public Health* 2021, 18, 432. [CrossRef]
- Jiao, X.; Kumar, R.; Billot, J.; Smith, R. Developing research skills and capability in higher education: Combining collaborative research with mentoring. J. Educ. Leadersh. Policy Pract. 2011, 26, 42–55.
- 22. Atkins, K.; Dougan, B.M.; Dromgold-Sermen, M.S.; Potter, H.; Sathy, V.; Panter, A.T. "Looking at Myself in the Future": How mentoring shapes scientific identity for STEM students from underrepresented groups. *Int. J. STEM Educ.* 2020, 7, 42. [CrossRef]
- 23. Case, D.O. Looking for Information: A Survey of Research on Information Seeking, Needs and Behavior, 2nd ed.; Elsevier: Amsterdam, The Netherlands, 2007.
- 24. Lopatto, D. The essential features of undergraduate research. Counc. Undergrad. Res. Q. 2003, 2, 139–142.
- Russell, S.H.; Hancock, M.P.; McCullough, J. Benefits of undergraduate research experiences. *Science* 2007, *316*, 548–549. [CrossRef]
 Blaney, J.M.; Kang, J.; Wofford, A.M.; Feldon, D.F. Mentoring relationships between doctoral students and postdocs in the lab
- sciences. *Stud. Grad. Postdr. Educ.* 2020, *11*, 263–279. [CrossRef]
 27. Chlomoudis, C.; Konstantinou, A.; Kostagiolas, P.; Pallis, P. Information needs and information-seeking behaviour of maritime students: A systematic literature review using the PRISMA method. *Libr. Manag.* 2022, *43*, 353–369. [CrossRef]
- 28. Moore, P. Information problem solving: A wider view of library skills. Contemp. Educ. Psychol. 1995, 20, 1–31. [CrossRef]
- Ellström, P.E. The many meanings of occupational competence and qualification. J. Eur. Ind. Train. 1997, 21, 266–273. [CrossRef]
- Adler, P.S. Skill trends under capitalism and the socialisation of production. In *The Skills that Matter;* Warhurst, C., Grugulis, I., Keep, E., Eds.; Palgrave Macmillan: Houndmills, UK, 2004.
- Brown, P.; Green, A.; Lauder, H. High Skills: Globalization, Competitiveness and Skill Formation; Oxford University Press: Oxford, UK, 2001.
- 32. Dixit, R.; Sinha, V. Role of Global Institutions in Competency Development. In *Handbook of Research on Global Institutional Roles for Inclusive Development*; IGI Global: Hershey, PA, USA, 2022; pp. 285–302.
- Forrier, A.; Sels, L. Temporary employment and employability: Training opportunities and efforts of temporary and permanent employees in Belgium. Work Employ. Soc. 2003, 17, 641–666. [CrossRef]
- 34. Marsick, V.J.; Watkins, K. Informal and Incidental Learning in the Workplace (Routledge Revivals); Routledge: London, UK, 2015.
- 35. Marsick, V.J.; Volpe, M.; Watkins, K.E. Theory and practice of informal learning in the knowledge era. *Adv. Dev. Hum. Resour.* **1999**, *1*, 80–95. [CrossRef]
- 36. Dickson, J.; Kirkpatrick-Husk, K.; Kendall, D.; Longabaugh, J.; Patel, A.; Scielzo, S. Untangling protégé self-reports of mentoring functions: Further meta-analytic understanding. *J. Career Dev.* **2014**, *41*, 263–281. [CrossRef]

- Rumjaun, A.; Narod, F. Social learning theory—Albert Bandura. In Science Education in Theory and Practice: An Introductory Guide to Learning Theory; Akpan, B., Kennedy, T.J., Eds.; Springer Cham: Cham, Switzerland, 2020; pp. 85–99.
- Edinyang, S.D. The significance of social learning theories in the teaching of social studies education. *Int. J. Sociol. Anthropol. Res.* 2016, 2, 40–45.
- 39. Ahn, J.N.; Hu, D.; Vega, M. "Do as I do, not as I say": Using social learning theory to unpack the impact of role models on students' outcomes in education. *Soc. Personal. Psychol. Compass* **2020**, *14*, e12517. [CrossRef]
- 40. Bandura, A. Principles of Behavior Modification; Holt, Rinehart & Winston: New York, NY, USA, 1969.
- 41. Bandura, A. Social Learning Theory; General Learning Press: New York, NY, USA, 1971.
- 42. Bandura, A. Social Learning Theory; Prentice-Hall: Englewood Cliffs, NJ, USA, 1977.
- 43. Bhatnagar, V.; Diaz, S.; Bucur, P.A. The need for more mentorship in medical school. Cureus 2020, 12, e7984. [CrossRef]
- 44. Speer, J.E.; Lyon, M.; Johnson, J. Gains and losses in virtual mentorship: A descriptive case study of undergraduate mentees and graduate mentors in STEM research during the COVID-19 pandemic. *CBE-Life Sci. Educ.* **2021**, *20*, 14. [CrossRef] [PubMed]
- 45. Luna, G.; Cullen, D.L. *Empowering the Faculty: Mentoring Redirected and Renewed*; ASHE-ERIC Higher Education Report No. 3.; ERIC Clearinghouse on Higher Education: Washington, DC, USA, 1995; pp. 20036–21183.
- Lorenzetti, D.L.; Shipton, L.; Nowell, L.; Jacobsen, M.; Lorenzetti, L.; Clancy, T.; Paolucci, E.O. A systematic review of graduate student peer mentorship in academia. *Mentor. Tutoring Partnersh. Learn.* 2019, 27, 549–576. [CrossRef]
- 47. Haeger, H.; Fresquez, C. Mentoring for inclusion: The impact of mentoring on undergraduate researchers in the sciences. *CBE-Life Sci. Educ.* **2016**, *15*, 36. [CrossRef] [PubMed]
- Agricola, B.T.; Prins, F.J.; van der Schaaf, M.F.; van Tartwijk, J. Teachers' diagnosis of students' research skills during the mentoring of the undergraduate thesis. *Mentor. Tutoring Partnersh. Learn.* 2018, 26, 542–562. [CrossRef]
- 49. Nolan, J.R.; McConville, K.S.; Addona, V.; Tintle, N.L.; Pearl, D.K. Mentoring undergraduate research in statistics: Reaping the benefits and overcoming the barriers. *J. Stat. Educ.* **2020**, *28*, 140–153. [CrossRef]
- 50. Laursen, S.; Hunter, A.B.; Seymour, E.; Thiry, H.; Melton, G. *Undergraduate Research in the Sciences: Engaging Students in Real Science*; John Wiley & Sons: New York, NY, USA, 2010.
- Malcolm, S.; Feder, M. (Eds.) Barriers and Opportunities for 2-Year and 4-Year STEM Degrees: Systemic Change to Support Students' Diverse Pathways; The National Academies Press: Washington, DC, USA, 2016.
- 52. Nelson Le-Gall, S.N.-L. Help-seeking: An understudied problem-solving skill in children. Dev. Rev. 1981, 1, 224–246. [CrossRef]
- 53. Nelson Le-Gall, S. Chapter 2: Help-seeking behavior in learning. Rev. Res. Educ. 1985, 12, 55–90.
- Ames, R.; Lau, S. An attributional analysis of student help-seeking in academic settings. J. Educ. Psychol. 1982, 74, 414–423. [CrossRef]
- 55. Aristoteles; Rini, P.S.; Wijanarko, N.A. The relationship of academic help-seeking with student achievement on nursing students in STIKes Muhammadiyah Palembang. *Enfermería Clínica* **2020**, *30*, 106–109. [CrossRef]
- 56. Dueñas, J.M.; Camarero-Figuerola, M.; Castarlenas, E. Academic help-seeking attitudes, and their relationship with emotional variables. *Sustainability* **2021**, *13*, 6120. [CrossRef]
- 57. Bates, M.J. Information behavior. In *Encyclopedia of Library and Information Sciences*, 4th ed.; CRC Press: Boca Raton, FL, USA, 2017; pp. 2074–2085.
- Case, D.O.; Given, L.M. Looking for Information: A Survey of Research on Information Seeking, Needs, and Behavior, 4th ed.; Emerald Group Publishing Limited: Bingley, UK, 2016.
- Association of College & Research Libraries. Information Literacy Competency Standards for Higher Education. 2000. Available online: http://www.ala.org/acrl/standards/informationliteracycompetency (accessed on 15 April 2023).
- Limberg, L.; Sundin, O.; Talja, S. Three theoretical perspectives on information literacy. *Hum. IT J. Inf. Technol. Stud. A Hum. Sci.* 2012, 11, 93–130.
- Lupton, M. Information Literacy and Learning [Doctoral Dissertation, Queensland University of Technology]. 2008. Available online: https://eprints.qut.edu.au/16665/1/Mandy_Lupton_Thesis.pdf (accessed on 15 April 2023).
- 62. Singh, K.P.; Kumar, M.; Khanchandani, V. Information needs and information seeking behavior of foreign students in University of Delhi: A survey. *Int. J. Knowl. Content Dev. Technol.* **2015**, *5*, 25–43. [CrossRef]
- 63. Wilson, T.D. Models in information behaviour research. J. Doc. 1999, 55, 249–270. [CrossRef]
- 64. Dahlqvist, C. Information-seeking behaviours of teacher students: A systematic review of quantitative methods literature. *Educ. Inf.* **2021**, *37*, 259–285. [CrossRef]
- 65. Stričević, I.; Rubinić, D. Librarians' assistance to students in the information seeking process: Perspectives of higher education teachers and librarians. *J. Acad. Librariansh.* **2023**, *49*, 102629. [CrossRef]
- Howlader, A.I.; Islam, M.A. Information-seeking behaviour of undergraduate students: A developing country perspective. *IFLA* J. 2019, 45, 140–156. [CrossRef]
- 67. Bălău, N.; Utz, S. Information sharing as strategic behaviour: The role of information display, social motivation and time pressure. *Behav. Inf. Technol.* 2017, *36*, 589–605. [CrossRef]
- Cho, I.; Park, H.; Kim, J.K. The relationship between motivation and information sharing about products and services on Facebook. *Behav. Inf. Technol.* 2015, 34, 858–868. [CrossRef]
- 69. Barclay, D.; Higgins, C.; Thompson, R. The partial least squares (PLS) approach to casual modeling: Personal computer adoption and use as an Illustration. *Technol. Stud.* **1995**, *2*, 285–309.

- Sosik, J.J.; Kahai, S.S.; Piovoso, M.J. Silver bullet or voodoo statistics? A primer for using the partial least squares data analytic technique in group and organization research. *Group Organ. Manag.* 2009, 34, 5–36. [CrossRef]
- 71. Chin, W.W. The partial least squares approach to structural equation modeling. Mod. Methods Bus. Res. 1998, 295, 295–336.
- 72. Henseler, J.; Ringle, C.M.; Sinkovics, R.R. The use of partial least squares path modeling in international marketing. In *New Challenges to International Marketing*; Emerald Group Publishing Limited: Bingley, UK, 2009; pp. 277–319.
- 73. Beebe, K.R.; Pell, R.J.; Seasholtz, M.B. *Chemometrics a Practical Guide*; Wiley-Interscience Series on Laboratory Automation; Wiley: New York, NY, USA, 1998; Volume 4.
- 74. Hair, J.F.; Ringle, C.M.; Sarstedt, M. PLS-SEM: Indeed a silver bullet. J. Mark. Theory Pract. 2011, 19, 139–151. [CrossRef]
- 75. Hair, J.F., Jr.; Matthews, L.M.; Matthews, R.L.; Sarstedt, M. PLS-SEM or CB-SEM: Updated guidelines on which method to use. *Int. J. Multivar. Data Anal.* 2017, 1, 107–123. [CrossRef]
- 76. Falk, R.F.; Miller, N.B. A Primer for Soft Modeling; University of Akron Press: Akron, OH, USA, 1992.
- Hair, J.F., Jr.; Sarstedt, M.; Hopkins, L.; Kuppelwieser, V.G. Partial least squares structural equation modeling (PLS-SEM): An emerging tool in business research. *Eur. Bus. Rev.* 2014, 26, 106–121. [CrossRef]
- Henseler, J.; Ringle, C.M.; Sarstedt, M. A new criterion for assessing discriminant validity in variance-based structural equation modeling. J. Acad. Mark. Sci. 2015, 43, 115–135. [CrossRef]
- Fornell, C.; Bookstein, F.L. Two structural equation models: LISREL and PLS applied to consumer exit-voice theory. *J. Mark. Res.* 1982, 19, 440–452. [CrossRef]
- 80. Hair, J.F., Jr.; Hult, G.T.M.; Ringle, C.; Sarstedt, M. A Primer on Partial Least Squares Structural Equation Modeling (PLS-SEM), 2nd ed.; Sage Publications: Thousand Oaks, CA, USA, 2016.
- 81. Urbach, N.; Ahlemann, F. Structural equation modeling in information systems research using partial least squares. J. Inf. Technol. Theory Appl. 2010, 11, 2.
- 82. Soliman, W.S.M.K. Investigating the effect of corporate governance on audit quality and its impact on investment efficiency. *Invest. Manag. Financ. Innov.* **2020**, *17*, 175–188.
- Fornell, C.; Larcker, D.F. Evaluating Structural Equation Models with Unobservable Variables and Measurement Error. J. Mark. Res. 1981, 18, 39–50. [CrossRef]
- 84. Nunnally, J.C.; Bernstein, I.H. Psychometric Theory, 3rd ed.; McGraw-Hill: New York, NY, USA, 1994.
- Ursachi, G.; Horodnic, I.A.; Zait, A. How reliable are measurement scales? External factors with indirect influence on reliability estimators. *Procedia Econ. Financ.* 2015, 20, 679–686. [CrossRef]
- 86. Hu, L.T.; Bentler, P.M. Fit indices in covariance structure modeling: Sensitivity to underparameterized model misspecification. *Psychol. Methods* **1998**, *3*, 424–453. [CrossRef]
- Henseler, J.; Dijkstra, T.K.; Sarstedt, M.; Ringle, C.M.; Diamantopoulos, A.; Straub, D.W.; Ketchen, D.J., Jr.; Hair, J.F.; Hult, G.T.M.; Calantone, R.J. Common Beliefs and Reality about Partial Least Squares: Comments on Rönkkö & Evermann. Organ. Res. Methods 2014, 17, 182–209.
- Bentler, P.M.; Bonett, D.G. Significance tests and goodness-of-fit in the analysis of covariance structures. *Psychol. Bull.* 1980, 88, 588–600. [CrossRef]
- Sarstedt, M.; Ringle, C.M.; Smith, D.; Reams, R.; Hair Jr, J.F. Partial least squares structural equation modeling (PLS-SEM): A useful tool for family business researchers. J. Fam. Bus. Strat. 2014, 5, 105–115. [CrossRef]
- 90. Cohen, J. Statistical Power Analysis for the Behavioral Sciences; Lawrence Erlbaum: Hillsdale, NJ, USA, 1988.
- 91. Cohen, J. A power primer. *Psychol. Bull.* **1992**, *112*, 155–159. [CrossRef]
- 92. Kock, N. Advanced mediating effects tests, multi-group analyses, and measurement model assessments in PLS-based SEM. *Int. J. e-Collab.* **2014**, *10*, 1–13. [CrossRef]
- Zhao, X.; Lynch Jr, J.G.; Chen, Q. Reconsidering Baron and Kenny: Myths and truths about mediation analysis. J. Consum. Res. 2010, 37, 197–206. [CrossRef]
- 94. Hair, J.F., Jr.; Hult, G.T.M.; Ringle, C.M.; Sarstedt, M. A Primer on Partial Least Squares Structural Equation Modeling (PLS-SEM), 3rd ed.; Sage Publications: Thousand Oaks, CA, USA, 2022.
- Younas, M.; Bari, M.W. The relationship between talent management practices and retention of generation 'Y' employees: Mediating role of competency development. *Econ. Res.-Ekon. Istraživanja* 2020, 33, 1330–1353. [CrossRef]
- Hong, A.J.; Kim, H.J. College students' digital readiness for academic engagement (DRAE) scale: Scale development and validation. *Asia-Pac. Educ. Res.* 2018, 27, 303–312. [CrossRef]
- Ipanaqué-Zapata, M.; Figueroa-Quiñones, J.; Bazalar-Palacios, J.; Arhuis-Inca, W.; Quiñones-Negrete, M.; Villarreal-Zegarra, D. Research skills for university students' thesis in E-learning: Scale development and validation in Peru. *Heliyon* 2023, 9, e13770. [CrossRef]

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.